Author's response to reviews

Title: Preconception care for diabetic women for improving maternal and fetal outcomes: a systematic review and meta-analysis

Authors:

Hayfaa A Wahabi (umlena@yahoo.com)
Rasmeia A Alziedan (ras_zeidan@hotmail.com)
Ghada A Bawazeer (gbawazeer@gmail.com)
Lubna A Al-Ansary (lansary@yahoo.com)
Samia A Esmaiel (drsoso123@hotmail.com)

Version: 8 Date: 14 October 2010

Author's response to reviews: see over
Dear Editorial Production Team:

The following changes have made to the manuscript as requested:

1. The authors list has been written as in the submission system in both style and order and the preferred style given has been followed.

2. The title page is written as requested.

3. Abstracted started in a new page.

4. Shading of tables has been removed.

5. Tables are added to the main manuscript with titles at the top and legend at the bottom.

6. Each figure starts and ends in one page and corresponding number to the figure has been added to the manuscript text.

7. A section for additional files has been added to the manuscript with title and description of content.

8. Title in both manuscript and system has been changed to sentence case as requested.

9. Introduction has been renamed as background in the manuscript.

10. We did not know how to change the citation of reference 4 because this is the citation recommended by the report editors on the following link:


11. The same applies to reference 7 as this is the citation retrieved by the software Reference Manager from Pubmed.

12. Vertical lines were removed from tables.

13. Figures are cropped as close as possible and titles were excluded from the image.
Hayfaa A A Wahabi,  
FRCOG, FRCS (Ed.), MMed Sci (A.R.T),  
MSC. Med-Educ.  
Consultant Obstetrician & Gynecologist,  
Assistant Professor EBHC & Knowledge Translation,  
King Saud University,  
PO Box 102599,  
Riyadh 11685,  
KSA.  
E-mail umlena@yahoo.com  Tel 00966553294955  Fax 0096614927987